Author's response to reviews

Title: Further delineation of Loeys-Dietz type IV syndrome in a family with mild vascular involvement and a TGFB2 splicing mutation

Authors:

Marco Ritelli (ritelli@med.unibs.it)
Nicola Chiarelli (chiarelli@med.unibs.it)
Chiara Dordoni (chiara.dordoni@hotmail.it)
Stefano Quinzani (stefano.quinzani@med.unibs.it)
Marina Venturini (marina.venturini@med.unibs.it)
Roberto Maroldi (maroldi@med.unibs.it)
Piergiacomo Calzavara-Pinton (calzavar@med.unibs.it)
Marina Colombi (colombi@med.unibs.it)

Version: 2
Date: 12 March 2014

Author's response to reviews:

Dear Editor,

thank you for your response to our MS: 1673467268123181 entitled “Further delineation of Loeys-Dietz type IV syndrome in a family with mild vascular involvement and a TGFB2 splicing mutation” by Marco Ritelli, Nicola Chiarelli, Chiara Dordoni, Stefano Quinzani, Marina Venturini, Roberto Maroldi, Piergiacomo Calzavara-Pinton and Marina Colombi.

We would be pleased to proceed with it as a Case Report in BMC Medical Genetics, as suggested.

We revised the manuscript and included an ethics statement, as requested. In particular, in the patients section we added that this study was approved by the medical ethical committee of the University Hospital Spedali Civili of Brescia and that the Declaration of Helsinki Principles were respected. Furthermore, we slightly modified the Abstract and the main text according to the guidelines of Case Reports.

Best regards,
Marina Colombi

Prof. Marina Colombi, PhD
Division of Biology and Genetics
Department of Molecular and Translational Medicine
University of Brescia, Viale Europa 11, 25123 Brescia, Italy
Phone: +39 030 3717265
Fax: +39 030 3701157
e-mail: colombi@med.unibs.it